Author's response to reviews

Title: Molecular testing for adult type Alport syndrome

Authors:

Genevieve Pont-Kingdon (pontkig@aruplab.com)
Kelli Sumner (kelli.sumner@aruplab.com)
Friederike Gedge (Friederike.gedge@aruplab.com)
Chris Miller (MILLERC@aruplab.com)
Joyce Denison (koyce.denison@m.cc.utah.edu)
Martin Gregory (martin.gregory@hsc.utah.edu)
Elaine Lyon (lyone@aruplab.com)

Version: 4 Date: 4 November 2009

Author's response to reviews:

list of authors have been corrected on the manuscript.
Each authors sent written approval to final version.
Title was formatted
Verticale lines removed from Tables
Few typographical and/or gramatical errors were corrected